Author's response to reviews

Title: Office-based screening for domestic violence: use of a child safety questionnaire.

Authors:

Dr Richard A Wahl (rwahl@peds.arizona.edu)
Doris J Sisk (doris@peds.arizona.edu)
Thomas M Ball (tball@u.arizona.edu)

Version: 5 Date: 16 May 2004

May 15, 2004

To the editors:

I am now returning my copyedited and formatted manuscript with all requested revisions included.

Thank you again for your continued assistance in the preparation of this document.

Manuscript modifications and responses to editorial queries:

Page 1:
Correct e-mail address for Thomas Ball: tball@u.arizona.edu

Page 3:
Last paragraph (Questionnaire design and use)
Wording "(designed for parents of infant, preschool-age, school-age, and adolescent patients)"
changed to match wording in abstract.

Page 4:
Last paragraph "ok to insert respectively?" - of course.

Page 7:
Authors' contributions revised as requested.

Page 8:
Reference 6: "Check page number"-- This is an electronic publication of the American Academy of Pediatrics. The Internet address of the article was added to the reference:
http://pediatrics.aappublications.org/cgi/content/full/108/3/e43

Page 9:
All tables reformatted as requested.

Reference 10: "What is town?" - Town name is College Station, in Texas, thus College Station, TX.

Sincerely,

Richard Wahl, M.D.
University of Arizona
Tucson, Arizona
USA